Author's response to reviews

Title: Bias in trials comparing paired continuous tests can cause researchers to choose the wrong screening modality

Authors:

Deborah H Glueck (Deborah.Glueck@ucdenver.edu)
Molly M Lamb (hgl9@cdc.gov)
Colin I Odonnell (Colin.Odonnell@ucdenver.edu)
Brandy M Ringham (Brandi.Ringham@ucdenver.edu)
John T Brinton (John.Brinton@ucdenver.edu)
Keith E Muller (Keith.Muller@biostat.ufl.edu)
John M Lewin (jlewinski@divrad.com)
Todd A Alonzo (talonzo@childrensoncologygroup.org)
Etta D Pisano (etpisano@med.unc.edu)

Version: 3 Date: 16 December 2008

Author’s response to reviews: see over